Author’s response to reviews

Title: Childbirth care in Egypt: A repeat cross-sectional analysis using Demographic and Health Surveys between 1995 and 2014 examining use of care, provider mix and immediate postpartum care content

Authors:

Miguel Pugliese-Garcia (Miguel.Pugliese-Garcia@lshtm.ac.uk)

Emma Radovich (Emma.Radovich@lshtm.ac.uk)

Oona MR Campbell (Oona.Campbell@lshtm.ac.uk)

Nevine Hassanein (nvnhassanein@yahoo.com)

Karima Khalil (karimahkhalil@gmail.com)

Lenka Benova (lbenova@itg.be)

Version: 3 Date: 09 Jan 2020

Author’s response to reviews:

Dear Drs Hickey and Coore,

Thank you very much for quick, useful feedback to finalise the manuscript.

Please, kindly find below a point-by-point response for the reviewed manuscript. We provide a detailed response to each reviewer/editorial point raised, describing what amendments have been made to the manuscript text and where these can be found.

Kind regards,
Miguel Pugliese and colleagues

Editor:

1. Thank you for stating the source for the map depicted in figure 1. As it has been provided by another source please acknowledge the source in the figure legend, including all relevant permissions given to use and adapt it.

Answer:

Thank you very much for highlighting this aspect on the map. As requested, the source used to create the map and the relevant license to reuse the data are now acknowledged in the figure legend.

Editor:

2. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Answer:
Thank you, we have proofread the article and we uploaded a revised submission for publication. We confirm we have removed tracked changes and highlighting and included only a single clean copy.